\documentclass[11pt]{article}
\usepackage{graphicx}
\usepackage[left=1in,right=1in,top=0.75in,bottom=0.5in]{geometry}
\usepackage{url}
\usepackage{amsmath}
\usepackage{upquote}
\pagestyle{myheadings}
\markright{MAT 4370: Programming Assignment 3}

\long\def\symbolfootnote[#1]#2{\begingroup%
\def\thefootnote{\fnsymbol{footnote}}\footnote[#1]{#2}\endgroup} 

\begin{document}
\thispagestyle{empty}
\begin{center}
  {\Large MAT 4370: Programming Assignment 3}\\
  Due: Wednesday, January 30
\end{center}

\section*{Problems}
All exercises for this assignment are from the King textbook.  Use the
following naming convention:  problem $n$ from chapter $k$
should be saved in the file named \verb!ex!$k$\verb!-!$n$\verb!.c!.   
\begin{itemize}
  \item Page~97, \#9. Your program should include a function to compare two given dates.  There
    are three possible outcomes, depending on whether the first date
    is less than, equal to, or greater than the second date. 

  \item Page~97, \#11. This problem provides practice with C's
    \verb!switch! statement.

  \item Page~124, \#10.  Reuse your function from exercise~9 above. 
\end{itemize}


\section*{What to Submit}
When your programs compile cleanly and work to your satisfaction,
place them in a folder named {\tt hw03}.  (Source code only please; no
other files need to be submitted.) Drag this
folder onto the SUBMIT icon, found in the Applications folder.


\end{document}
